Author’s response to reviews

Title: Cultural Safety Strategies for Rural Indigenous Palliative Care: A Scoping Review

Authors:

Kaela Schill (kschill@ucalgary.ca)

Susana Caxaj (susana.caxaj@ubc.ca)

Version: 2 Date: 25 Nov 2018

Author’s response to reviews:

To whom it may concern:

Thank you for the feedback. We believe we have reviewed and addressed all comments and believe this has strengthened the paper overall. Due to the extent of the revisions, please see the supplementary attachment (2018 11 24_Palliative Care Revisions Table) for detailed responses to the reviewer and editor comments. Changes have been made in the manuscript (titled: Manuscript revisions) in track changes.

Warm regards,

Kaela Schill